
diff -r 494b44b439bf -r 2f9fe225ecc8 handouts/ho07.tex
--- a/handouts/ho07.tex	Thu Dec 31 23:40:48 2015 +0000
+++ b/handouts/ho07.tex	Tue Jan 05 01:46:41 2016 +0000
@@ -633,7 +633,7 @@
 \lstinputlisting[language=JVMIS]{../progs/test-small.j}
 \caption{Generated code for a test program. This code can be 
 processed by an Java assembler producing a class-file, which
-can be run by the \pcode{java}-program.\label{test}}
+can be run by the {\tt{}java}-program.\label{test}}
 \end{figure}
 
 \end{document}




